Author's response to reviews

Title: Extending the spectrum of Ellis van Creveld syndrome: a large family with a mild mutation in the EVC gene

Authors:

- Hakan Ulucan (ulucanh@mail.nih.gov)
- Davut Gül (davutgul2001@yahoo.com)
- Julie C Sapp (sappj@mail.nih.gov)
- John Cockerham (jcockerh@cnmc.org)
- Jennifer J Johnston (jjohnsto@mail.nih.gov)
- Leslie G Biesecker (leslieb@helix.nih.gov)

Version: 7 Date: 20 October 2008

Author's response to reviews:

Last check and typographical corrections have been done